UCSF UC San Francisco Previously Published Works

Title

Concordance of ompA types in children re-infected with ocular Chlamydia trachomatis following mass azithromycin treatment for trachoma

Permalink

https://escholarship.org/uc/item/605225zs

Journal PLOS Neglected Tropical Diseases, 16(3)

ISSN

1935-2727

Authors

Mosenia, Arman Chin, Stephanie A Alemayehu, Wondu <u>et al.</u>

Publication Date

2022

DOI

10.1371/journal.pntd.0010237

Copyright Information

This work is made available under the terms of a Creative Commons Attribution License, available at <u>https://creativecommons.org/licenses/by/4.0/</u>

Peer reviewed

Check for updates

Citation: Mosenia A, Chin SA, Alemayehu W, Melese M, Lakew T, Zhou Z, et al. (2022) Concordance of *ompA* types in children re-infected with ocular *Chlamydia trachomatis* following mass azithromycin treatment for trachoma. PLoS Negl Trop Dis 16(3): e0010237. https://doi.org/10.1371/ journal.pntd.0010237

Editor: Jeremiah M. Ngondi, RTI International, UNITED REPUBLIC OF TANZANIA

Received: September 28, 2021

Accepted: February 7, 2022

Published: March 28, 2022

Copyright: © 2022 Mosenia et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This work received funding from the International Trachoma Initiative, the Bernard Osher Foundation, That Man May See, the Peierls Foundation, the Bodri Foundation, the Harper Inglis Trust, the South Asia Research Fund, Research to Prevent Blindness (TML). The study was also supported by funding from National Institutes of RESEARCH ARTICLE

Concordance of *ompA* types in children reinfected with ocular *Chlamydia trachomatis* following mass azithromycin treatment for trachoma

Arman Mosenia^{1,2}, Stephanie A. Chin¹, Wondu Alemayehu³, Muluken Melese³, Takele Lakew³, Zhaoxia Zhou¹, Thuy Doan^{1,4}, Vicky Cevallos¹, Thomas M. Lietman^{1,4,5}, Jeremy D. Keenan^{1,4}*

Francis I. Proctor Foundation, University of California, San Francisco, California, United States of America,
School of Medicine, University of California, San Francisco, California, United States of America,
Orbis International, New York, New York, United States of America,
Department of Ophthalmology, University of California, San Francisco, California,
San Francisco, California, United States of America,
Department of Epidemiology & Biostatistics, University of California, San Francisco, California, United States of America,

* jeremy.keenan@ucsf.edu

Abstract

Background

The chlamydial major outer membrane protein, encoded by the *ompA* gene, is a primary target for chlamydial vaccine research. However, human studies of *ompA*-specific immunity are limited, and prior studies have been limited in differentiating re-infection from persistent infection. The purpose of this study was to assess whether children living in trachomaendemic communities with re-infections of ocular chlamydia were more likely to be infected with a different or similar genovar.

Methodology and findings

The study included 21 communities from a trachoma-hyperendemic area of Ethiopia that had been treated with a mass azithromycin distribution for trachoma. Conjunctival swabbing was offered to all children younger than 5 years of age at baseline (i.e., pre-treatment), and then at follow-up visits 2 and 6 months later. Swabs were subjected to polymerase chain reaction (PCR) to detect *C. trachomatis*. A random sample of 359 PCR-positive swabs, stratified by study visit and study community, was chosen for *ompA* sequencing. In addition, *ompA* sequencing was performed on all swabs of 24 children who experienced chlamydial re-infection (i.e., positive chlamydial test before treatment, negative test 2 months following mass distribution of azithromycin, and again a positive test 6 months post-treatment). *ompA* sequencing was successful for 351 of 359 swabs of the random sample and 44 of 48 swabs of the re-infection sample. In the random sample, *ompA* types clustered within households more than would be expected by chance. Among the 21 re-infected children with complete *ompA* data, 14 had the same *ompA* type before and after treatment.

Health [U10EY016214 (TML), R21AI55752 (TML), K23EY019071 (JDK)]. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Conclusion

The high frequency of *ompA* concordance suggests incomplete genovar-specific protective immunity and the need for multiple antigens as vaccine targets.

Author summary

Trachoma is the world's leading infectious cause of blindness. Periodic mass azithromycin distributions have not resulted in elimination. Re-infection is thought to be an important barrier to elimination, but studies of serovar- or genovar-specific immunity and re-infection with ocular *Chlamydia trachomatis* have been limited. The present study investigated chlamydial genotypes in children who were infected with ocular chlamydia before azithromycin treatment and were confirmed to have cleared the infection 2 months later, but then were again infected at 6 months. The majority of re-infections were caused by the same *ompA* strain that was carried pre-treatment, which suggests poor protective immunity to chlamydia. The *ompA* types clustered by household, suggesting members of the same household as a source of re-infection. These findings could inform future mass treatment strategies and vaccine development.

Introduction

The major outer membrane protein (MOMP), coded by the *ompA* gene, is an immunodominant antigen of human *Chlamydia trachomatis* infections, and a primary target for vaccine development. [1] Chlamydial serovar classification is based on the MOMP, with serovars A-C associated with trachoma, D-K with urogenital infections, and L1-L3 with lymphogranuloma venereum. Animal studies suggest that natural protective immunity mediated through MOMP may be serovar-specific, since experimental re-infection with the same serovar causes less clinical inflammation than the primary infection, whereas re-infection with a different serovar produces a similar degree of inflammation as the first infection. [2] Studies of genital infections in humans have found some supportive evidence of serovar-specific immunity, since re-infections have usually been found to be caused by different serovars than the initial infection. [3,4] However, existing studies are limited by relatively low sample sizes, challenges in differentiating re-infection from persistent infection, and variable study timepoints.

A cluster-randomized trial of mass azithromycin distribution for trachoma provided an opportunity to further assess the genovar-specificity of natural protective immunity in *C. trachomatis.* [5,6] The trial was conducted in an area with hyperendemic trachoma, and thus reinfection after antibiotic treatment was relatively frequent. Participants were likely to be exposed to the same strains of ocular chlamydia before and after treatment given the geospatial clustering observed in trachoma. [7] Conjunctival swabs were collected on all pre-school children at standardized timepoints before and after azithromycin treatment, regardless of symptoms. A sample of swabs underwent sequencing for the *ompA* gene, which codes for the immunodominant antigen of *C. trachomatis.* We assessed the likelihood that children with an infection at baseline who were subsequently re-infected had the same *ompA* type during both infections. We hypothesized a lack of genotype-specific protective immunity, which would result in a relatively high proportion of re-infections with the same *ompA* type as the original infection.

Methods

Ethics statement

Ethical approval was obtained from the University of California, San Francisco and the National Ethical Clearance Committee of the Ethiopian Science and Technology Commission. Due to high illiteracy in the study area, guardians provided verbal informed consent for all study activities.

Study design

This is a secondary observational analysis of data that was collected for a cluster-randomized trial set in the Gurage zone of Ethiopia from 2003–2006. [5,6] In the trial, 40 communities had a baseline enumerative census and monitoring visit, and then received mass azithromycin treatment of their entire population (20mg/kg for children, 1 g for adults) and no further treatments until after the month 6 monitoring visit. All children aged 1-5 years were offered conjunctival swabbing for detection of C. trachomatis with the AMPLICOR polymerase chain reaction (PCR) assay (Roche Diagnostics USA, Indianapolis, Indiana) at baseline and then 2 and 6 months following antibiotic treatment. The 21 communities with the highest prevalence of infection at month 2 were selected for the present study (Fig 1). All 21 communities had received mass azithromycin distribution at baseline, with a mean antibiotic coverage of 89% (95% confidence interval [CI] 86–92%) in the 20 communities with available coverage data. Coverage data was lost for 1 community known to have been treated with mass azithromycin. All swabs with positive chlamydial PCR results from month 2 were selected for ompA sequencing, as was a random sample of the same number of swabs per community from the baseline and month 6 visits (i.e., a stratified random sample). The *ompA* sequences from this random sample have been reported previously. [8,9] In addition, swabs from all children with a positive C. trachomatis result at baseline, negative result at month 2, and positive result at month 6 were selected for *ompA* sequencing. Methods for polymerase chain reaction amplification and sequencing of the ompA gene region have been reported previously. [8] A single nucleotide polymorphism was considered a different ompA genotype. ompA types were labeled with genovar (A or B) and an arbitrary number (i.e., A1 to A11 and B1 to B4).

Statistical analyses

All households in which more than 1 child contributed an *ompA* result were included in a permutational multivariate analysis of variance (PERMANOVA) test to assess whether dissimilarity in *ompA* types was greater between households than within households (child-level analysis; Euclidean distance matrix of binary *ompA* indicator variables; dissimilarity partitioned over the nested hierarchical classes of community and household; 10,000 repetitions). Children with a positive *C. trachomatis* result at baseline, negative result at 2 months, and positive result at 6 months were considered to have cleared infection after antibiotic treatment and been subsequently re-infected. These children were included in a permutation test to assess whether the observed within-person concordance of the baseline and month 6 *ompA* types was higher than would be expected by chance. A significance level of 0.025 was set for each analysis. Analyses and plots were done in R version 4 (R for Statistical Computing, Vienna Austria).

Results

A total of 1556 children aged 1–5 years were enumerated in the baseline census from the 21 study communities. Conjunctival swabs were collected from 1393 children at baseline, 1386





https://doi.org/10.1371/journal.pntd.0010237.g001

children at month 2, and 1362 children at month 6 (Fig 2). The mean community-level prevalence of *C. trachomatis* among 1–5 year-olds was 56% (95%CI 49–64%) before treatment, 9% (95%CI 7–11%) 2 months after the antibiotic distribution, and 12% (95%CI 9–15%) 6 months after antibiotics. [8]

The stratified random sample consisted of 122 swabs from baseline, 122 swabs from month 2, and 115 swabs from month 6, with equal numbers of swabs from each community at each study timepoint except for 4 communities with fewer PCR-positive results at month 6. Of 359 swabs selected as part of the random sample across the 3 timepoints, *ompA* sequencing was successful for 351 (98%). The nucleotide sequences of each of these *ompA* types have been reported elsewhere; the present study follows the same numbering system as the previous report. [8] A total of 15 different *ompA* types were identified, ranging from A1 to A11 and from B1 to B4. Each *ompA* type had at least one amino acid difference relative to the other sequences. [8] The most common *ompA* types were A1, which accounted for about half of swabs (N = 64 at baseline, N = 57 at month 2, and N = 57 at month 6), and B1, which accounted for almost 15% of swabs (N = 18 at baseline, N = 16 at month 2, and N = 16 at month 6). Data on *ompA* type were contributed by a total of 274 children (mean age 3.2 years; 48% female) from 237 households, with 36 households contributing more than 1 child. Of these 36 households, 35 contained only a single *ompA* type across all children and/or



https://doi.org/10.1371/journal.pntd.0010237.g002

timepoints (Fig 3A). The *ompA* types of a given community were statistically significantly more similar within the same household than between different households (PERMANOVA P<0.001).

A total of 24 children had a swab positive for *C. trachomatis* at baseline, negative at 2 months, and then positive again at 6 months, and 21 of these children had successful *ompA* sequencing results at both the baseline and 6-month visits. Of these 21 children, 14 (67%, 95% CI 43–85%) had the same strain at both timepoints, which was significantly more than would be expected by chance (permutation P<0.001; Fig 3B). In 3 households, two siblings experienced re-infections. In each case, pairs of siblings shared the same *ompA* type at each timepoint, with two of the pairs having the same *ompA* type.

Discussion

It is well established that children living in areas with hyperendemic trachoma are likely to experience repeated infections with ocular chlamydia, but the serovars or genovars of re-infections have not been well described. [6] This study found that two-thirds of children re-infected with ocular strains of *C. trachomatis* after mass azithromycin treatment were re-infected with the same *ompA* type. The *ompA* types of children living in the same household were more similar than the *ompA* types of children living in different households of the same community, which is compatible with the theory that household members or nearby neighbors are the most likely sources of re-infection. These findings are consistent with a smaller study of

А										E	3				
		A1	-	- 1		4.0	A1	A1			Î	4	A4	-	A1
	1	+	B2	B2		10	+	A1	A1			1	A4	_	A1
		+	_	A1			A1 A1 +				A1	_	A2		
		+	-	A1			+	A1	_			2	A2	—	A7
		+	A6	A6			I	A1	A1	ī į			A3	_	A2
		+	A6]		11		A1	+			3	A1		A1
		-	A1	A1			B2	B2	_			5	A1	_	B1
	2	_	-	A1			-		B2		6	_	A1	_	A1
	-		B2	B2			+	B2	B2			6	A3	_	A3
			B2	B2			+	B2	+				A3		A3
	3	A1	-				A1	-					A1		A1
		+	A1	A1				A1	-			+	_	+	
			A1				+	+ + A1	A1			8	+	_	A1 D1
	4 5 6	D1	AT	+			AT		AT		÷	0			
		Ы		DI		12	-		_			9	Λ2 Λ1		Α2 Λ1
			ы	Δ1			+	A1 +	Δ1					_	
			_	A1			A3	A3	A3			12	A4	_	A4
		_	A4	+		13 14 15 17 18 19	+	A3	A3		13		B1	_	B1
			_	A4				B2	B2			13	A2		B2
		B1	B1	+					B2			14	A1	_	A1
			B1	+			A1	A1	A1		15	45	A1	_	A1
		B1	_	-				_	A1			15	A1		A1
				B1			+	A1	+			20	A2	_	A2
		B1	-	-			+		A1				1	1	1
		_	B1	B1			A1	-	A1						
		+	B1	B1			+		A1						
		+	B1	+			+	A1	A1						
		+	A1	A1			+	A1	+						
		+	_	A1				B1	B1						
		A3	-	-			+	BI	BI						
		A3							AI						
	7			_ 				AI		8					
		Δ2	_												
		A2	_	_											1
		+	A2	_											
		+	A2	_											
	-	+		A2											
	9	+		A2											
-		0	2	6			0	2	6				0	2	6
	Month					Month					Month				

Fig 3. Concordance of *ompA* **type in 21 Ethiopian communities.** Panel A shows all households in the random sample that had multiple children with *ompA* results. Each row of boxes represents a unique child's *ompA* results over time, from before the mass azithromycin distribution (i.e., month 0) to 6 months post-treatment. Blank spaces represent visits with no specimen collected. Shaded boxes represent those swabs selected for the study, and hence those eligible for *ompA* sequencing. Shaded boxes are labeled with the sequenced *ompA* type using the arbitrary naming conventions of this study (i.e., A1, A2, B1, etc.) or with "+" if the *ompA* type could not be determined. Unshaded boxes are labeled according to the polymerase chain reaction (PCR) results for *C. trachomatis* as positive ("+") or negative ("-"). Households are delineated with thick black lines and communities with grey dashed lines. Panel B shows all children experiencing *Chlamydia trachomatis* re-infection after mass azithromycin treatment (i.e., positive test results at baseline, negative results at month 2, and positive results again at month 6), regardless of whether they were selected for the random sample. Discordant *ompA* types are shown in red.

https://doi.org/10.1371/journal.pntd.0010237.g003

Egyptian children with trachoma that found clustering of *ompA* type at the household-level (N = 34 households) and re-infection with the same *ompA* type at the individual level (N = 3 children) [10]. The results confirm that natural protective immunity to chlamydia is incomplete and suggest that chlamydia vaccines relying on MOMP alone may be less effective than those relying on multiple antigenic epitopes.

Studies of sexually transmitted chlamydia infections have also documented re-infections with the same *ompA* type, but at lower proportions than observed in the present study. For example, a prospective study of sex workers in Kenya found that 15 of 40 (38%) women with two positive tests for C. trachomatis more than 4 weeks apart had the same ompA type at both tests. [4] In a retrospective study of 74 patients from an Indiana clinic with two positive chlamydia tests, 88% of whom had an intervening negative chlamydia culture, 25 (34%) had the same *ompA* type at both visits. [11] A prospective study of 35 patients from a clinic in Alabama with positive tests for C. trachomatis before and 6 months after azithromycin therapy found that 7 (20%) had the same *ompA* type at both timepoints. [3] However, the generalizability of these studies is unclear. Two studies did not document a negative test between the two positive tests, making it difficult to differentiate re-infection from persistent infection. [3,4] Two studies assessed for re-infections at variable timepoints [4,11]. Most re-infections from the Kenya study occurred in women with HIV, which may have resulted in altered natural immunity. Moreover, people re-infected with genital C. trachomatis may be exposed to more diverse set of serovars and genovars than children with trachoma, who are likely exposed to the same *ompA* types in circulation in the community. [12]

The study has several strengths, including its population-based sampling, standardized timepoints, strict definition of re-infection (i.e., requirement of a negative post-treatment chlamydial test) and setting in a place with a high chance of being re-exposed to the same *ompA* type after antibiotic treatment. The study's biggest limitation is its focus on a single antigenic epitope, the MOMP. While MOMP is probably the most important immunodominant epitope and the target of much vaccine research, more extensive sequencing could have provided information about other immunogenic epitopes.

In summary, two-thirds of children who successfully cleared ocular *C. trachomatis* infection after mass azithromycin treatment but were subsequently re-infected frequently had the same *ompA* type during the original infection and the re-infection. *ompA* types clustered by house-hold over the duration of the study, suggesting fellow household members could be potential sources of re-infection. Chlamydia vaccines targeting only the MOMP may be less likely to confer protective immunity than vaccines containing multiple antigenic epitopes.

Supporting information

S1 Data. *ompA* **types in random sample.** A stratified random sample of children aged 1–5 years from 21 communities in the Gurage Zone, Ethiopia who had a positive polymerase chain

reaction (PCR) test for *Chlamydia trachomatis* underwent sequencing for the ompA gene. Separate stratified random samples were selected before mass azithromycin treatment, 2 months after treatment, and 6 months after treatment. (CSV)

S2 Data. *ompA* types in re-infection sample. All children aged 1–5 years from 21 communities in the Gurage Zone, Ethiopia who had polymerase chain reaction (PCR) evidence of ocular *Chlamydia trachomatis* infection before mass azithromycin treatment, then a negative PCR test 2 months after treatment, and then again a positive PCR test 6 months after treatment underwent *ompA* sequencing.

(CSV)

Acknowledgments

We would like to thank the Ethiopian Ministry of Health and the many health care professionals who helped us organize and implement our fieldwork in Ethiopia, including Tadesse Kebede, Berhanu Fikre, Mifta Shifa and Tadesse Birru.

Author Contributions

Conceptualization: Stephanie A. Chin, Wondu Alemayehu, Muluken Melese, Takele Lakew, Vicky Cevallos, Thomas M. Lietman, Jeremy D. Keenan.

Data curation: Arman Mosenia, Zhaoxia Zhou.

Formal analysis: Arman Mosenia, Thuy Doan, Thomas M. Lietman, Jeremy D. Keenan.

Funding acquisition: Thomas M. Lietman.

Investigation: Stephanie A. Chin, Wondu Alemayehu, Muluken Melese, Takele Lakew, Vicky Cevallos, Thomas M. Lietman.

Methodology: Stephanie A. Chin, Vicky Cevallos, Thomas M. Lietman.

Supervision: Vicky Cevallos, Thomas M. Lietman, Jeremy D. Keenan.

Writing - original draft: Arman Mosenia.

Writing – review & editing: Arman Mosenia, Stephanie A. Chin, Wondu Alemayehu, Muluken Melese, Takele Lakew, Zhaoxia Zhou, Thuy Doan, Vicky Cevallos, Thomas M. Lietman, Jeremy D. Keenan.

References

- Poston TB, Gottlieb SL, Darville T. Status of vaccine research and development of vaccines for Chlamydia trachomatis infection. Vaccine. 2019 Nov 28; 37(50):7289–94. https://doi.org/10.1016/j.vaccine. 2017.01.023 PMID: 28111145
- Taylor HR. Development of Immunity to Ocular Chlamydial Infection. Am J Trop Med Hyg. 1990 Apr 1; 42(4):358–64. https://doi.org/10.4269/ajtmh.1990.42.358 PMID: 2331044
- Kapil R, Press CG, Hwang ML, Brown L, Geisler WM. Investigating the Epidemiology of Repeat Chlamydia trachomatis Detection after Treatment by Using C. trachomatis OmpA Genotyping. Bourbeau P, editor. J Clin Microbiol. 2015 Feb; 53(2):546–9. https://doi.org/10.1128/JCM.02483-14 PMID: 25472488
- Brunham RC, Kimani J, Bwayo J, Maitha G, Maclean I, Yang C, et al. The Epidemiology of Chlamydia trachomatis within a Sexually Transmitted Diseases Core Group. J Infect Dis. 1996 Apr 1; 173(4):950– 6. https://doi.org/10.1093/infdis/173.4.950 PMID: 8603976

- Chidambaram JD, Alemayehu W, Melese M, Lakew T, Yi E, House J, et al. Effect of a Single Mass Antibiotic Distribution on the Prevalence of Infectious Trachoma. JAMA. 2006 Mar 8; 295(10):1142. <u>https:// doi.org/10.1001/jama.295.10.1142</u> PMID: 16522834
- Lakew T, House J, Hong KC, Yi E, Alemayehu W, Melese M, et al. Reduction and Return of Infectious Trachoma in Severely Affected Communities in Ethiopia. Ko AI, editor. PLoS Negl Trop Dis. 2009 Feb 10; 3(2):e376. https://doi.org/10.1371/journal.pntd.0000376 PMID: 19190781
- Altherr FM, Nute AW, Zerihun M, Sata E, Stewart AEP, Gessese D, et al. Associations between Water, Sanitation and Hygiene (WASH) and trachoma clustering at aggregate spatial scales, Amhara, Ethiopia. Parasit Vectors. 2019 Dec; 12(1):540. <u>https://doi.org/10.1186/s13071-019-3790-3</u> PMID: 31727155
- Chin SA, Morberg DP, Alemayehu W, Melese M, Lakew T, Chen MC, et al. Diversity of Chlamydia trachomatis in Trachoma-Hyperendemic Communities Treated With Azithromycin. Am J Epidemiol. 2018 Sep 1; 187(9):1840–5. https://doi.org/10.1093/aje/kwy071 PMID: 29617922
- Chin SA, Alemayehu W, Melese M, Lakew T, Cevallos V, Lietman TM, et al. Association of Chlamydia trachomatis ompA genovar with trachoma phenotypes. Eye. 2018 Aug; 32(8):1411–20. https://doi.org/ 10.1038/s41433-018-0069-4 PMID: 29503448
- Bain DL, Lietman T, Rasmussen S, Kalman S, Fan J, Lammel C, et al. Chlamydial Genovar Distribution after Communitywide Antibiotic Treatment. J Infect Dis. 2001 Dec 15; 184(12):1581–8. https://doi.org/ 10.1086/324661 PMID: 11740734
- Batteiger BE, Fraiz J, Newhall V WJ, Katz BP, Jones RB. Association of Recurrent Chlamydial Infection with Gonorrhea. J Infect Dis. 1989 Apr 1; 159(4):661–9. https://doi.org/10.1093/infdis/159.4.661 PMID: 2926160
- Hayes LJ, Pecharatana S, Bailey RL, Hampton TJ, Pickett MA, Mabey DCW, et al. Extent and Kinetics of Genetic Change in the ompl Gene of Chlamydia trachomatis in Two Villages with Endemic Trachoma. J Infect Dis. 1995 Jul; 172(1):268–72. https://doi.org/10.1093/infdis/172.1.268 PMID: 7797929